Author's response to reviews

Title: Training auscultatory skills: computer simulated heart sounds or additional bedside training? A randomized trial on third-year medical students

Authors:

Øystein Sverdrup (oystein.sverdrup@studmed.uio.no)
Torstein Jensen (torstein.jensen@lds.no)
Svein Solheim (svein.solheim@ulleval.no)
Knut Gjesdal (knut.gjesdal@medisin.uio.no)

Version: 5 Date: 14 January 2010

Author's response to reviews:

There was an undesired space above last line